Author's response to reviews

Title: Health and health care utilisation among asylum seekers and refugees in the Netherlands: design of a study

Authors:

Dr Annette AAM Gerritsen (aam.gerritsen@vumc.nl)
Inge I Bramsen (i.bramsen@vumc.nl)
Dr Walter W Deville (w.deville@nivel.nl)
Loes LHM van Willigen (ivanwil@xs4all.nl)
Hans JE Hovens (hanhov@deltabouman.nl)
Henk HM van der Ploeg (hm.vanderploeg@vumc.nl)

Version: 2 Date: 3 Mar 2004

Dear Emma Parkin,

The following changes have been made:
1. The authors qualifications were removed form the title page.
2. The authors were written in the order in which they are listed in the submission system.
3. Information about funding was written in the Acknowledgements section.

I hope I informed you well,
Best regards, on behalf of Dr. Annette Gerritsen,

Femke van der Linden
(research-assistent)